Author’s response to reviews

Title: Protective and restorative effects of the traditional Chinese medicine Jitai tablet against methamphetamine-induced dopaminergic neurotoxicity

Authors:

Shasha Xu (shasha.1218@163.com)
Shaoang Tu (tushaoang@mail.bnu.edu.cn)
Jinlong Gao (gaojinlong06@mail.bnu.edu.cn)
Jia Liu (liujiadan@mail.bnu.edu.cn)
Zhirui Guo (gzr12358@163.com)
Jinming Zhang (zhangjm@sina.com)
Xingdang Liu (xingdliu@yahoo.com)
Jianhui Liang (liangjh@bjmu.edu.cn)
Yiyun Huang (henry.huang@yale.edu)
Mei Han (hanmei@bnu.edu.cn)

Version: 4 Date: 14 Jan 2018

Author’s response to reviews:

Dear Editor:

Thank you very much for inviting us to revise our manuscript. We have carried out the following revisions in the revised manuscript according to the editor’s kind suggestions.

1. We have revised the abstract to only include the following subheadings:

   - Background
   - Methods
   - Results
   - Conclusions
2. We have included the email addresses of all authors on the title page of this revised manuscript.

3. We have provided a List of abbreviations after the Conclusions section. The abbreviations used in the text have been defined in the text at first use and included in this list.

4. We have deleted the supplementary file, "Checklist S1.docx".

    We hope that our revisions are sufficient for publication in BMC Complementary & Alternative Medicine.

    Thank you very much for your consideration.

Sincerely yours

Mei HAN Ph.D. PI
Professor of Medicinal Chemistry
Key Lab of Radiopharmaceutical Chemistry, Ministry of Education
College of Chemistry, Beijing Normal University
No.19 Xin Jie Kou Wai Avenue, Hai Dian District
Beijing 100875, P.R. China
Tel: +86-10-62207786
Email: hanmei@bnu.edu